Author’s response to reviews

Title: The Epidemiology of HIV and other Sexually Transmitted Infections in African, Caribbean and Black Men in Toronto, Canada

Authors:

LaRon Nelson (laron.nelson@yale.edu)

Wangari Tharao (wangari@whiwh.com)

Winston Husbands (winston.husbands@sympatico.ca)

Ting Sa (ting.sa@cchmc.org)

Nanhua Zhang (nanhua.zhang@cchmc.org)

Sameer Kushwaha (sameer.kushwaha@mail.utoronto.ca)

David Absalom (AbsalomD@SMH.CA)

Rupert Kaul (rupert.kaul@utoronto.ca)

Version: 4 Date: 14 Mar 2019

Author’s response to reviews:

Thank you for the key comments on the final version of the manuscript. We wish to clarify that we did not enroll any participants under the age of 16; therefore, there were no children involved in our study. We realized that Table 1 included an age range of "15-24" which may have given the incorrect impression that some participants may have been as young as 15. We had two participants who were age 16. No participants were younger than 16. We amended the lower age bracket in Table 1 to "16-24" to minimize any risk of misinterpretation because we did not have any participants younger than 16.

We are happy to resubmit the final clean version of the manuscript and tables.

Sincerely,

LaRon E. Nelson, PhD, RN, FNP, FNAP, FAAN

Associate Dean for Global Health & Equity
Independence Foundation Professor and Associate Professor

Yale University